Author’s response to reviews

Title: Fluctuations in the Incidence of Type 1 Diabetes in the United States from 2001-2015: A Longitudinal Study

Authors:

Mary Rogers (maryroge@umich.edu)
Catherine Kim (cathkim@med.umich.edu)
Tania Banerjee (tania@med.umich.edu)
Joyce Lee (joyclee@med.umich.edu)

Version: 2 Date: 11 Oct 2017

Author’s response to reviews:

Reviewer #5: I tend to agree with the authors' explanations for not using covariate adjusted age-specific incidence rate in both Table 1 and Figures 1-5. However, I would suggest to summarize these explanations at somewhere in either Method or Discussion section; simply for justification.

Response: Thank you for the suggestion. These explanations are now summarized in the Methods Section (lines 158-159 on page 7).